

The Pearson Institute Discussion Paper No. 35

# **Transnational insurgents**

# Evidence from Colombia's FARC at the border with Chávez's Venezuela

Luis R. Martínez

2016

Working paper



# **Transnational Insurgents:** Evidence from Colombia's FARC at the border with Chávez's Venezuela

Luis R. Martínez \*

THIS VERSION: SEPTEMBER 2016 FIRST VERSION: AUGUST 2013

#### Abstract

Many insurgent groups operate across international borders, but transnational insurgent activities are seldom observable and little is known about their consequences. In this paper, I show that activity by Colombian insurgent group FARC increased disproportionately in Colombian municipalities next to the border with Venezuela after Hugo Chávez became president of the latter. This finding is consistent with increased FARC presence in Venezuela during the Chávez administration, given that the military and transport technologies employed by the insurgents severely constrained the area in which a safe haven across the border allowed them to expand their operations. Exploiting the plausible exogeneity of the change of government in Venezuela, I find that municipalities more exposed to a cross-border guerrilla sanctuary experience large increases in the intensity of violence, as well as reductions in local tax revenue and educational enrolment.

Keywords: civil conflict, foreign influence, irregular warfare, FARC, Hugo Chávez JEL Classification: D740, F590

<sup>&</sup>lt;sup>\*</sup>Harris School of Public Policy, University of Chicago. luismartinez@uchicago.edu

# 1. Introduction

Recent research on internal conflict has increasingly explored its international dimension.<sup>1</sup> However, insurgent groups' ability to operate outside of their target state has received little scholarly attention. This is somewhat surprising, given that as many as 55 % of insurgencies active after 1945 have operated across international borders (Cunningham et al., 2007). Some well-documented historical examples include the presence of Palestinian rebels in Jordan and Lebanon in the 1970s, of the Nicaraguan Contras in Honduras and Costa Rica in the 1980s and of Rwandan Hutu militias in what at the time was Zaire in the 1990s (Byman et al., 2001; Salehyan, 2009). These examples, two of which led to costly inter-state military confrontations (the 1982 Lebanon war and the first Congo war, respectively), highlight the threat that "transnational" insurgent activities pose for conflict resolution and development.<sup>2</sup>

One major obstacle for research on transnational insurgency has been the limited availability of information on this topic. Some host nations welcome foreign insurgent groups into their territory, but often keep this sponsorship secret to avoid diplomatic or military reprisals (Byman et al., 2001; Salehyan, 2010). Other insurgent groups impose their presence on weak neighboring states, but the low state capacity that these groups exploit also limits data availability and compromises its quality. Despite these difficulties, the "Non-

<sup>&</sup>lt;sup>1</sup>For example, Hegre and Sambanis (2006) and Gleditsch (2007) document robust corelations between the onset of internal conflict and political and economic conditions in neighbouring countries, while Gleditsch et al. (2008) show a positive correlation between internal conflict and international disputes. Buhaug and Gates (2002) report that internal conflicts taking place near an international border tend to have a larger geographical scope. Refugee flows as a mechanism through which conflict propagates across borders have been studied by Salehyan and Gleditsch (2006) and Salehyan (2008b). Doyle and Sambanis (2000) find that foreign intervention, in the form of multilateral peacebuilding operations, is correlated with less violence, while Regan (2002) shows that third-party intervention has a positive correlation with conflict duration. A more recent strand of literature has systematically uncovered the detrimental effects of superpower intervention on political stability and conflict, particularly in the context of the cold war (Kalyvas and Balcells, 2010; Dube et al., 2011; Berger et al., 2013a,b; Albornoz and Hauk, 2014; Nunn and Qian, 2014).

<sup>&</sup>lt;sup>2</sup>The term "transnational" has been used in other contexts to refer to internal conflicts that draw significant numbers of foreign fighters (Malet, 2013) or to refer to terrorist groups with activities in more than one target state (Rosendorff and Sandler, 2005).

State Actor" dataset constructed by Cunningham et al. (2007) includes some information on the extraterritorial presence of many insurgent groups.<sup>3</sup> Using that data, Idean Salehyan (2007, 2008a) has shown that conflicts involving transnational insurgents tend to last longer and are positively correlated with the probability of war between the target and host states, but standard issues of endogeneity prevent a causal interpretation of these findings.

The present paper furthers our understanding of transnational insurgency by employing sub-national data from the target state to provide evidence on cross-border rebel activity and its effects on local levels of violence and living conditions. For this purpose, I exploit the change of government in a neighboring country as a source of plausibly exogenous variation in rebels' access to foreign territory. The setting is the border between conflict-ridden Colombia and Venezuela, where left-wing candidate Hugo Chávez took office as president in 1999. The change of government in Venezuela affected the dynamics of the Colombian conflict, as Chávez (who would remain in power until his death in 2013) never hid his sympathy for Colombia's leftist insurgencies, FARC and ELN.<sup>4</sup> Evidence from various sources, discussed in detail below, indicates that Chávez had close ties with these insurgent groups and that their presence in Venezuela increased during his administration.

The availability of a safe haven across an international border reduces the cost and increases the probability of success for various insurgent activities, allowing transnational rebels to expand their operations. However, the Colombian guerrillas' ability to exploit a cross-border refuge was not homogeneous throughout their target state. In particular, the insurgents' short-range weaponry and scarce use of mechanized transport limited their increased operational ability to the immediate vicinity of the border with Venezuela. As a result, Colombian municipalities at the border with Venezuela were the ones exposed to the effects of increased guerrilla access to the other side of the border.

<sup>&</sup>lt;sup>3</sup>For example, variables "rebpresosts" and "presname" indicate for dyad 237 (FARC vs. Colombian government) that there is "some" FARC presence in "Venezuela."

<sup>&</sup>lt;sup>4</sup>FARC is the the acronym for "Fuerzas Armadas Revolucionarias de Colombia" (Revolutionary Armed Forces of Colombia). ELN stands for "Ejército de Liberación Nacional" (National Liberation Army).

Focusing on FARC, the largest guerrilla group, I show that the intensity of its activities increased disproportionately in Colombian municipalities located at the border with Venezuela after Chávez took office in 1999. Figure 1 illustrates this result with the raw data. It displays the yearly average of FARC Events and the corresponding 95% confidence interval for border municipalities and for all others. The graph shows that the intensity of insurgent activity was identical in the two groups up to 1998, but diverged from 1999 onwards, increasing disproportionately at the border.<sup>5</sup>

Regression results, including municipality and region-year fixed effects, confirm that FARC activity was 0.3 standard deviations higher in border municipalities than in the rest of the country during the Chávez administration. This is a large increase in insurgent activity and corresponds to 1.16 extra FARC events per 10,000 inhabitants, relative to a sample mean of 1.11. It is also quite a costly increase, since it mainly involves acts of terrorism, town sieges, terrestrial piracy and armed clashes with government forces. I further show that, as a result, border municipalities had higher homicide rates, collected less local taxes (property and business tax) and had lower educational enrolment rates.<sup>6</sup>

These results are robust to (i) the inclusion of a broad set of control variables for the time-varying effect of various municipality characteristics; (ii) the employment of different measures of proximity to the border; (iii) the use of different datasets on the Colombian conflict; (iv) the estimation of models for count data; and (v) the addition of further controls for both the paramilitary expansion and for the increase in U.S. military aid that roughly coincided with the start of the Chávez administration. I also provide evidence against alternative explanations based on changes in economic or political conditions specific to the border region.

Additional exercises reveal two other interesting results. First, activity by

<sup>&</sup>lt;sup>5</sup>The difference diminishes substantially between 2002 and 2004, but reappears in the following years. This pattern matches existing evidence on the distancing between Chávez and FARC that took place during the period of heightened political instability in Venezuela.

<sup>&</sup>lt;sup>6</sup>Rohner et al. (2013) employ a similar research design that exploits distance to an international border as a source of plausibly exogenous variation in exposure to conflict. See also Miguel and Roland (2011); Dube et al. (2013); Caselli et al. (2015); Serneels and Verpoorten (2015).

ELN, the other main insurgent group present in Colombia, also increased disproportionately at the border with Venezuela in the early years of the Chávez administration but declined as FARC expanded into some of its main areas of operation. Secondly, FARC activity increased disproportionately in municipalities at the border with Ecuador after 2003, when FARC commander Raúl Reyes is known to have relocated across the border, but not after 2007, when Chávez's political ally, Rafael Correa, became president.<sup>7</sup>

The remainder of the paper is structured as follows. Section 2 provides some background information on the Colombian conflict and Venezuela's involvement. Section 3 presents the theoretical framework. Section 4 introduces the data. In section 5, I discuss the empirical strategy. Section 6 presents the main results, robustness checks and evidence against alternative explanations. Finally, section 7 concludes.

# 2. Background

#### 2.1. A Recent History of the Colombian Conflict

Left-wing insurgent groups have been trying to overthrow the Colombian government since 1964. FARC is the largest guerrilla group in the country and had as many as 20,000 operatives at its peak around 2002 (Boraz, 2007; El Tiempo, 2012; Dube and Vargas, 2013). The other main insurgent group, ELN, is much smaller and has never had more than 5,000 operatives (Boraz, 2007). Neither rebel group has ever had the military capacity nor the popular support necessary to succeed in its political ambitions. However, the state's limited presence in rural areas allowed the insurgents to find a stable source of income in the extortion of local businessmen, at the same time as they carried out guerrilla warfare amid favourable geographic conditions (Rangel, 2000; Semana, 2005; Pizarro, 2007; Palacios, 2012; Richani, 2013). The result

<sup>&</sup>lt;sup>7</sup>This finding is consistent with existing evidence on FARC's ability to recurrently exploit weak patrolling of the border by Ecuador and on its inability to obtain protection from the Ecuadorian government (ICG, 2004; IISS, 2011a,b).

was a long-lasting but relatively low-intensity armed conflict.<sup>8</sup>

Throughout most of the conflict, FARC activity was concentrated in a handful of sparsely-populated jungle and mountain areas. But in the 1990s the organization became increasingly able to raise revenue from the various stages of drug processing and trafficking as coca cultivation soared in Colombia (Boraz, 2007; Angrist and Kugler, 2008; Otis, 2014). The additional resources allowed FARC to expand throughout the country, particularly towards the south and the east, as the maps in Figure 4 illustrate. ELN, on the other hand, was at a financial disadvantage due to its limited involvement in the drug business and only remained active in its traditional strongholds, mainly near the border with Venezuela (El Espectador, 2014).

A series of military successes for FARC in the mid-1990s led president Andrés Pastrana (1998 - 2002) to hold peace negotiations with the group in a demilitarised zone (DMZ) comprising 5 municipalities in the departments of Meta and Caquetá (See panel (a) of Figure 3). The peace talks took place simultaneously with the dramatic escalation of the conflict shown in Figure 2. FARC kept expanding in the south and the east and increasingly moved into ELN territories, further adding to the latter's decline (El Tiempo, 2005a,b; IISS, 2011a; Avila, 2012).

The collapse of Pastrana's peace process a few months before the 2002 presidential elections contributed to Alvaro Uribe's victory. Uribe was elected (and re-elected in 2006) with a clear mandate to fight the guerrillas (Cortés et al., 2012; Rozo, 2016). Uribe's counterinsurgency strategy profited from a U.S. military aid package worth more than \$1 billion and known as "Plan Colombia" (Dube and Naidu, 2015). The resulting increase in military activity, shown in Figure 2, allowed government forces to deal a series of major blows to FARC, including the death of several of its top leaders and the rescue of its most high-profile hostages (Fergusson et al., 2014).

It was also during the Uribe administration that the paramilitary organi-

 $<sup>^{8}</sup>$ In the well-known UCDP/PRIO armed conflict dataset the Colombian conflict has always surpassed the 25 battle-death threshold since 1964, but only in eight years did it surpass the threshold of 1,000 battle deaths.

zation AUC demobilized.<sup>9</sup> The first paramilitary groups were created in the early 1980s by land owners and drug lords who decided to organize private armies in response to extortion from the guerrillas (Acemoglu et al., 2013). These groups survived thanks to their involvement with the drugs trade and to contributions from businesses who paid for security in their areas of operation. In 1997 many of these groups united as the AUC, with the explicit purpose of defeating the guerrillas. Between 1999 and 2004 the paramilitaries expanded significantly, as documented in Figure 2. It is estimated that they had around 15,000 combatants at their peak in 2003 (Dube and Vargas, 2013).

A new round of peace talks between FARC and the Colombian government of Juan Manuel Santos (2010-2018), Uribe's successor, started in 2012. An agreement for FARC to lay down their weapons was reached in 2016.

#### 2.2. Venezuela and the Colombian Conflict

The presence of Colombian insurgent groups at the border with Venezuela dates back at least to the 1980s, when occasional skirmishes between ELN and Venezuelan security forces were not infrequent. FARC's presence at the border was weaker than ELN's at the time but also less visible, partly because of the group's "border policy" of not carrying out military operations in foreign territory (Semana, 1995; Ramírez, 2003; IISS, 2011a). To address the threat from the Colombian guerrillas, Venezuelan president Rafael Caldera (1994-1998) created two new military units to patrol the border, curtailed the local population's civil rights and provided military tribunals with legal authority in the area (El Tiempo, 1995a, 1997). Caldera allowed Colombian troops into Venezuela when in hot pursuit of insurgents and demanded the same from Colombia (El Tiempo, 1995b, 1996, 1998; Avila, 2012). Still, by the time of the Venezuelan presidential elections of December 1998, six of FARC's 'fronts' were active at the border (Boraz, 2007; IISS, 2011a).<sup>10</sup>

 $<sup>^{9}\</sup>mathrm{AUC}$  stands for "Autodefensas Unidas de Colombia" (United Self-defense Groups of Colombia).

<sup>&</sup>lt;sup>10</sup>Each front has between 100 and 500 operatives. Each front belongs to one of seven 'blocs', each led by a member of FARC's top decision-making body, the Secretariat.

This election was won by Hugo Chávez, a former army lieutenant who had led a failed coup attempt in 1992 and was imprisoned until 1995. He allegedly received a contribution of \$130,000 from FARC during that time.<sup>11</sup> Shortly after taking office in February 1999, Chávez discontinued Caldera's border security policies. He declared Venezuela a neutral party in Colombia's armed conflict, breaking away from his country's long-standing support for the Colombian government (Ramírez, 2003). Chávez apparently thought of the Colombian guerrillas as potential allies in case of a U.S. invasion (Avila, 2012). News reports on Venezuela's supply of weapons and provision of sanctuary to the insurgents soon started appearing (El Tiempo, 1999, 2000a,b; El Universal, 2002; Semana, 2002; Robinson, 2003).

Following the approval of a new constitution in 2000, Chávez was re-elected for a six-year presidential term (2001-2006), but political polarization rose quickly afterwards. In April 2002 a failed coup attempt took place and in December of that same year employees from Venezuela's national oil company (PDVSA) went on a prolonged strike lasting over two months. Domestic political instability, a changing international environment following the 9/11 attacks and FARC's excesses inside Venezuela apparently prompted Chávez to moderate his relationship with the Colombian insurgencies in this period (El Tiempo, 2004b; IISS, 2011a; Avila, 2012). Bilateral cooperation resumed as the Venezuelan security forces proved willing to strike against the rebels (El Tiempo, 2004a,c; Boraz, 2007). But the illegal capture of a FARC spokesman in Caracas by Colombian security forces in December 2004 triggered a diplomatic crisis and apparently rekindled Chávez's relation with the insurgents (BBC News, 2004, 2005b; IISS, 2011a).

Political stability had recently returned to Venezuela following the failed recall referendum of August 2004 (Hsieh et al., 2011). The price of oil was on the rise and this allowed the Venezuelan government to pursue a more agressive foreign policy that would see friendly candidates win presidential elections in several Latin American countries over the next years (Corrales, 2009; Clem

 $<sup>^{11}</sup>$ See items I.22, I.773, I.782, I.2995 and I.2997 of the personal correspondence of FARC commander "Raúl Reyes" in IISS (2011b).

and Maingot, 2011). Chávez also implemented more agressive domestic policies under the banner of "21st Century Socialism" (Rodríguez, 2008; Ortega and Rodríguez, 2008; Grier and Maynard, 2016).

He was re-elected for a second six-year term (2007-2012) in 2006, as the Colombian government and FARC tried to agree terms for a prisoner exchange. In 2007, Chávez was designated as mediator for the process and publicly hosted senior FARC representatives at the presidential palace in Caracas (BBC News, 2007). He also allegedly held parallel private meetings with FARC representatives in which he offered to provide the group with \$300 million.<sup>12</sup> Following Chávez's dismissal as mediator by the Colombian government, FARC released several hostages as a friendly gesture towards him (The Washington Post, 2007; BBC News, 2008c). Chávez reciprocated during his annual address to the Venezuelan National Assembly in January 2008 when he stated:

"They are not terrorist groups, they are true armies that occupy territory in Colombia. [Political] recognition must be given to FARC and ELN, they are insurgent forces that have a political project, a 'bolivarian' project that we respect here" (El Tiempo, 2008, own translation).

In March 2008, the attack on a FARC camp in neighbouring Ecuador by Colombian government forces resulted in the death of the rebels' second-incommand, "Raúl Reyes". Colombia's violation of Ecuadorian sovereignty led to the suspension of diplomatic relations by the Venezuelan government, which also deployed troops at the border (BBC News, 2008b,d). Documents found in Reyes' laptop provide extensive evidence of the ties between FARC and the Chávez administration (IISS, 2011a,b). They also indicate that Reyes moved into Ecuador in 2003 after the suspension of the Pastrana peace talks led to the closure of the DMZ. This was well before the election of pro-Chávez candidate Rafael Correa as president of Ecuador in 2007. FARC apparently tried on repeated occasions to strike a deal regarding the safety of its operatives with

 $<sup>^{12}\</sup>mathrm{See}$  items I.2838, I.2850, I.2866, I.2890, I.2907 and I.2929 of the personal correspondence of "Raúl Reyes" in IISS (2011b).

the different governments of Ecuador since 2000 but were unsuccessful (ICG, 2004; IISS, 2011a).

In 2009, Colombian troops seized weapons at a FARC camp that had apparently been sold by a Swedish manufacturer to the Venezuelan Armed Forces (Semana, 2009; The New York Times, 2009a). Intercepted FARC communications provided additional evidence on Venezuela's role as an intermediary in arms deals involving the insurgents (The New York Times, 2009b). Coincidentally, Gonsalves et al. (2009) report in the account of their experience as FARC hostages that the tags on the military uniforms given to them read "Made in Venezuela." Chávez denied the allegations and imposed restrictions on bilateral trade.

At a summit of the Organisation of American States (OAS) in 2010, the Colombian government divulged satellite images of alleged FARC camps inside Venezuela (BBC News, 2010). Colombian intelligence reports estimated at the time that over 1,500 FARC operatives were distributed among 28 camps inside Venezuela (El Espectador, 2010a,b). Chávez again denied the allegations but also criticised FARC and invited them to demobilize, something he had sporadically done in the past (BBC News, 2008a; El Tiempo, 2010). Chávez was re-elected for a third six-year term (2013-2018) in 2012, but died from cancer in March 2013.

# 3. Theoretical Framework

Following the seminal work of Becker (1968), the academic literature has understood insurgent groups as rational actors who make decisions (such as whether to rebel or not, when and where to strike, etc.) after carefully weighing their costs and benefits (Kalyvas, 2006; Weinstein, 2007). From this perspective, we expect rebel groups to avoid actions that are likely to have a high cost (all else equal), such as the seizure of weapons or the loss of operatives. These negative outcomes not only reduce the insurgents' resources, but may allow counterinsurgents to acquire valuable information on the group's members and activities. They can also reduce group morale and popular support. Hence, a decrease in the probability of a negative outcome should lead to an increase in the overall level of insurgent activity.<sup>13</sup>

One way in which insurgent groups can reduce the riskiness of their actions is by moving their operations outside of their target state, as governments face potentially high costs for unauthorized counterinsurgency activities within another country's territory.<sup>14</sup> These costs range from economic and diplomatic sanctions to a heightened risk of inter-state war (Salehyan, 2008a). Thus, insurgent groups that are able to operate on foreign soil can potentially set themselves beyond the reach of the target state's military and judiciary.

Naturally, the extent to which an international border protects insurgents depends on the behaviour of the host state. In order to become transnational, a rebel group must find either a weak neighbor who is unable to repel it or a welcoming one who is unwilling to do so (Salehyan, 2007). The discussion above suggests that FARC's expansion into Venezuela and Ecuador provides good examples of 'welcoming' and 'weak' host nations, respectively.<sup>15</sup> More generally, Byman et al. (2001) report that 59% of insurgent groups active after the end of the cold war have relied critically on support from foreign states, often neighboring ones that have provided sanctuary.<sup>16</sup>

Access to foreign territory can have a profound effect in an irregular conflict, as it allows insurgents to better prepare and execute their actions (Lindsay, 1962). Across the border, rebels can safely train operatives until it is optimal to act and, by crossing back, they can evade opposition forces after an action has taken place. El Tiempo (2002) provides an example: on March 17 of 2002, FARC operatives entered the Colombian municipality of Tibú from Venezuela and burnt eleven vehicles on the main road. As troops from the

<sup>&</sup>lt;sup>13</sup>The equilibrium outcome will also depend on efforts in counterinsurgency, which may increase in response (Berman et al., 2011).

<sup>&</sup>lt;sup>14</sup>Salehyan (2009, p. 26) notes that "state boundaries are perhaps the most fundamental international institutions in the modern state system."

<sup>&</sup>lt;sup>15</sup>This characterization fails to account that armed groups also exploited the limited state control over the Colombia-Venezuela border (Boraz, 2007; Avila, 2012).

<sup>&</sup>lt;sup>16</sup>Salehyan et al. (2011) find that insurgent groups are more likely to receive foreign support if they are of intermediate strength, if they represent a supranational group and if the government they are fighting against is itself a recipient of foreign assistance.

Colombian Army approached, the insurgents crossed back into Venezuela, but re-entered Tibú three days later and engaged in combat with the Colombian military, with twelve insurgents and five soldiers dying as a result. The remaining rebels retreated back into Venezuela and attacked from across the border with gas cylinders filled with explosives. The following day the guerrillas attacked yet again and the ensuing fight led to the death of nine insurgents and seventeen soldiers.

The episode above illustrates the strategic advantage that insurgent groups obtain from access to foreign territory as well as the high human cost from additional insurgent activity. It also highlights how this advantage is circumscribed to targets located relatively near to the border. If the insurgents penetrate deeper into the target state, they are more likely to be attacked by government forces, both before and after the action has taken place. Thus, rebels' ability to exploit access to foreign territory has a limited geographic scope. This feature sets it apart from other forms of foreign influence, like the provision of weapons or funding, as those resources can be delivered and employed in areas far from the border.

Some characteristics of Colombian insurgencies particularly limit their ability to exploit a cross-border refuge. The guerrillas' arsenal consists mostly of short-range weapons, such as AK-47 rifles and 60 mm mortars, which have an effective range of no more than 5 or 6 km (Cragin and Hoffman, 2003; BBC News, 2005a; Gonsalves et al., 2009). The rebels also mostly refrain from using motorized vehicles or the existing road network due to the increased vulnerability to attacks. Former FARC hostages report moving mainly on foot, with thick forests and jungles providing cover from military aircraft (Gonsalves et al., 2009; Betancourt, 2011). Given that troops on foot can cover around 25 km per day on rough terrain (U.S. Army, 1990), it is unlikely that FARC could have exploited a safe haven in Venezuela to carry out activities in municipalities not located right at the border. On average, non-border municipalities that are immediate neighbors of border ones are 92 km away from the border line. Operating even in those municipalities would require the insurgents to go relatively deep into Colombian territory, giving up the strategic advantage provided by their cross-border refuge.

The key hypothesis that emerges from this discussion is that rebels' access to territory in a neighbouring country leads to an increase in their activities in the vicinity of the respective border. An additional hypothesis is that rebels' access to foreign territory has a heterogeneous effect across different types of insurgent activities.<sup>17</sup> Actions that require many operatives and have a high probability of retaliation, such as an attack on a military installation, may become increasingly attractive. Also those that require the transportation and sheltering of hostages or goods, such as kidnapping or road piracy. On the other hand, the rebels' ability to carry out actions that require few operatives and little time, like a murder, are less likely to be affected.

### 4. Data

I use sub-national panel data on the Colombian conflict from research centre CEDE at Los Andes University in Bogotá. CEDE collects information from official government sources and provides 19 different activity indicators for each non-state armed group (FARC, ELN, AUC) and nine activity indicators for the Colombian Armed Forces.<sup>18</sup> Data is available for 1,099 municipalities (98% of the total) from 1993 until 2008. Municipalities lacking CEDE conflict data are shown in grey in the maps in Figures 3 and 4. Following Acemoglu et al. (2013) and Camacho and Rodríguez (2013), I create an "Events" variable for each actor by adding all of its activity indicators in the CEDE data and dividing by 1993 population (Mueller, 2016).<sup>19</sup> Nine of the 19 indicators

<sup>&</sup>lt;sup>17</sup>I thank an anonymous referee for pointing this out. For more information on rebels' tactical choice, see Bueno de Mesquita (2013) and Wright (2016).

<sup>&</sup>lt;sup>18</sup>For non-state armed groups, these indicators are: terrorist act (explosive, incendiary, other), assault to private property, road block, terrestrial piracy, illegal checkpoint, armed contact, ambush, siege, incursion, attack against installation, massacre, political assassination (failed, succesful), murder of civilian, kidnap (politician, military, civilians). For government forces: arrest, explosive defusal, demobilization, destruction of cocaine laboratory, other anti-narcotic operation, raid, seizure of weapons, hostage release, hostage rescue.

<sup>&</sup>lt;sup>19</sup>For new municipalities created during the sample period I divide by initial population. The results in the paper are robust to the use of contemporary population figures instead of the 1993 values. They are also robust to the exclusion of new and broken-up municipalities.

for non-state armed groups take no non-zero values for several years, but the results are unaffected if I exclude them.<sup>20</sup> I also analyze separately each of the indicators of FARC activity that have at least one non-zero value every year.

To check the sensitivity of the results to the source of information on the conflict, I use the replication data from Dube and Vargas (2013), which I will refer to as the DV dataset. This data is produced by CERAC, a Colombian think-tank that collects information from national and local newspapers and complements it with reports from non-government organizations working in remote areas (Restrepo et al., 2004; Fergusson et al., 2014). The DV dataset includes the yearly number of attacks, massacres and political kidnappings by "Guerrillas" for 966 municipalities from 1988 to 2004. I create a "Guerrilla Events" variable by adding these three indicators and dividing again by 1993 population. For ease of comparison, all "Events" variables from both datasets are standardized.

For most of the analysis, I divide municipalities into three groups: municipalities that share a border with Venezuela ("border"), municipalities that do not share a border with Venezuela but are contiguous to those that do ("one-from-border") and municipalities that are none of the above. There are 41 municipalities at the border (from seven different departments) and 43 onefrom-border municipalities, out of a total of 1,123. These are shown in panel (a) of Figure 3. I classify municipalities near the border with Ecuador using analogous criteria.

To check the robustness of the results to the way in which proximity to Venezuela is measured, I calculate the great-circle distance from the border to the urban centre of each municipality (the "Cabecera municipal" where the seat of local government is located).<sup>21</sup> This is the appropriate metric in the

The estimates are available upon request.

<sup>&</sup>lt;sup>20</sup>The following indicators have no non-zero values from the year in brackets until the end of the sample in 2008: incendiary terrorist acts (2004), massacres (2004), incursion (2007), attack on installation (2007), political murder attempt (2004), political murder (2007), military kidnap (2004), political kidnap (2007). Additionally, the indicator on road blocks has all zeros before 2003. Results are also unaffected if I set 2003 as the final year of the sample.

<sup>&</sup>lt;sup>21</sup>The great-circle distance is the shortest distance between two points on a sphere. Similar metrics have often been used in previous work (Miguel and Roland, 2011; Rohner et al.,

present context, rather than the distance to the nearest border crossing or the driving distance (i.e. Dube et al. (2013); Dell (2015)) because Colombian guerrilla groups move mainly on foot and do not usually employ the road network. Panel (b) in Figure 3 shows the results.

Municipal data on fixed physical characteristics (area, altitude, distance to the nearest market), state presence (notary office, agricultural bank office, tax collection office, health center or hospital), homicides, educational enrolment and electoral results comes from CEDE (Acevedo and Bornacelly, 2014; Pachón and Sánchez, 2014). The Unmet Basic Needs (NBI) index and the population figures (total and urban/rural shares) are provided by the National Department of Statistics (DANE). DANE also provides information on the date of creation and the location of new municipalities.<sup>22</sup> Yearly information on local public finance (natural resource royalties, transfers from central government, local tax revenue) is provided by the National Department of Planning (DNP).

Table 1 shows summary statistics for the main variables employed in the paper.

## 5. Empirical Strategy

The main specification for the empirical analysis is the following:

$$y_{i,j,t} = \beta_1 [D(Chávez)_t \times D(Border Venezuela)_i] + \beta_2 [D(Chávez)_t \times D(One-from-border Venezuela)_i] + \alpha_i + \gamma X_{i,t} + \delta_{j,t} + \epsilon_{i,j,t}$$
(1)

where  $y_{i,j,t}$  is an outcome of interest (mainly FARC Events) in municipality *i* from region *j* in year *t*. D(Chávez)<sub>t</sub> is a dummy equal to one from 1999 onwards. D(Border Venezuela)<sub>i</sub> and D(One-from-border Venezuela)<sub>i</sub> are time-invariant

<sup>2013;</sup> Caselli et al., 2015; Serneels and Verpoorten, 2015).

 $<sup>^{22}69</sup>$  new municipalities were created between 1993 and 2008. They separated from 92 existing ones in 20 different departments

dummies created in accordance with the definitions above. The coefficients of interest,  $\beta_1$  and  $\beta_2$ , capture the difference-in-the-difference in the dependent variable between the corresponding municipalities and the rest of the country (non-border and non-one-from-border) after Chávez came to power, conditional on the controls that I list below.

 $\alpha_i$  is a municipality fixed effect. It accounts for persistent heterogeneity between municipalities near the border with Venezuela and the rest of the country.<sup>23</sup> Its inclusion is desirable because these municipalities are different on various observable characteristics, as can be seen in panel C of Table 1. The estimates could still be biased if the characteristics that are correlated with proximity to the border had time-varying effects that coincided with the Chávez years (e.g. higher conflict intensity after 1999 affecting disproportionately poorer or more rural municipalities). To address this concern, I include as controls  $(X_{i,t})$  a full set of year fixed effects interacted with each of the variables listed in panel C of Table 1.

 $\delta_{j,t}$  is a region-year fixed effect that controls for events that simultaneously affect the entire country, as well as those that only affect municipalities within a given region. For example, the region-year fixed effect captures the effect of changes over time in various forms of foreign influence, such as the provision of weapons or money to the insurgents, that have a large geographic scope. To construct the region-year fixed effect, I classify the 32 departments in the country in six 'regions', roughly corresponding to natural geographic divisions.<sup>24</sup> Low levels of within-department variation in proximity to the border with Venezuela prevent me from using a more conservative department-year

<sup>&</sup>lt;sup>23</sup>Previous research has shown that the intensity of civil conflict tends to be higher in border areas. For example, Buhaug and Rod (2006) find that proximity to the border was positively correlated to separatist conflict in Africa between 1970 and 2001. Michalopoulos and Papaioannou (2016) show that political violence is higher in the vicinity of international borders that divide the traditional homelands of African ethnic groups.

<sup>&</sup>lt;sup>24</sup>Andean: Cundinamarca, Boyacá, Santander, Norte de Santander; North-West: Cordoba, Antioquia, Chocó, Quindío, Risaralda, Caldas; Caribbean: Atlántico, Bolivar, Magdalena, Sucre, Cesar, La Guajira; South-West: Cauca, Valle, Nariño, Tolima, Huila; Eastern Plains: Meta, Casanare, Arauca, Vichada; South: Caquetá, Vaupés, Amazonas, Guainía, Putumayo, Guaviare.

fixed effect.<sup>25</sup>

The error term,  $\epsilon_{i,j,t}$ , is two-way clustered by municipality and departmentyear, following Cameron et al. (2011). Double clustering allows for arbitrary correlation of the residuals within a municipality across years and across a department for a given year. Thus, the two-way clustering accounts for serial correlation within a municipality and for spatial correlation within a department.

Equation (1) corresponds to a standard differences-in-differences research design. In the present context, the "parallel trends" identifying assumption implies that the difference in the intensity of FARC activity between the border area and the rest of the country would have remained unchanged after Chávez came to power in the absence of FARC's increased access to Venezuela. This assumption is plausible given the exogenous events that led to the change of government in Venezuela.<sup>26</sup> Below, I use data from the pre-Chávez period to formally test the parallel trends assumption.

### 6. Results

#### 6.1. Main Results

I begin the analysis by estimating a simplified version of equation (1) for FARC Events with only municipality and year fixed effects. The estimates in column 1 of Table 2 indicate that, on average, FARC activity grew 0.24 standard deviations (SD) more in border municipalities than in the rest of the country after Chávez came to power. FARC activity in one-from-border municipalities had a relative increase of almost the same magnitude, but the

 $<sup>^{25}</sup>$ The adjusted R-squared from a cross-sectional regression of a "border or one-fromborder" dummy on the six region dummies is 0.10, but rises to 0.65 if I replace them with department dummies. However, estimates with department-year fixed effects (not reported) show a 0.32 SD increase in FARC activity in municipalities at the border with Venezuela between 1999 and 2002 (p<0.001).

<sup>&</sup>lt;sup>26</sup>Chávez's electoral success was driven by popular discontent with the traditional political parties following years of sluggish economic performance (Corrales, 2013). Up to election day there was a large degree of uncertainty regarding the ability of the candidates from the traditional parties to catch up with Chávez (McCoy, 1999).

standard error is somewhat larger. The inclusion of controls for the timevarying effects of fixed municipal characteristics (column 2) leads to a slight decrease in the magnitude of the point estimate for border municipalities, but it remains statistically significant at the 10 % level. The point estimate for one-from-border municipalities decreases substantially.

Column 3 shows results from my preferred specification with municipality and region-year fixed effects, plus the additional controls. The results indicate that there was a 0.32 SD relative average increase in FARC activity in border municipalities when Chávez was in office.<sup>27</sup> Not only is this coefficient statistically significant at the 1% level, but it also represents a quite sizable effect: 1.16 extra FARC events per 10,000 inhabitants, relative to a sample mean of 1.11. The point estimate for one-from-border municipalities is further reduced by the additional controls. This result is consistent with the hypothesis that FARC's ability to exploit their increased access to Venezuela was limited to the immediate vicinity of the border and did not extend to one-from-border municipalities.

To further test this hypothesis, I exploit the more fine-grained variation in proximity to Venezuela provided by the distance to the border. I modify equation (1) by interacting the Chávez time indicator with four dummies that partition the distance to the border. The results in column 4 of Table 2 show that, as expected, the increase in insurgent activity decreases with distance to the border and disappears after 30 km. The mean distance to Venezuela from the urban center of border municipalities is 27 km, while from one-from-border municipalities it is 92 km.

Column 5 re-estimates the model from column 3 but with ELN Events as the dependent variable. The point estimate is positive for border municipalities and negative for one-from-border municipalities, but in both cases the magnitude is quite small and statistically insignificant. However, these average

<sup>&</sup>lt;sup>27</sup>The increase in the magnitude of the coefficient following the inclusion of region-year fixed effects reflects the intensification of the conflict that took place after 1999 in "control" areas far from the border with Venezuela, such as the coffee-growing regions of Caldas/Quindío in the west (Dube and Vargas, 2013) and the area near the border with Ecuador in the south-west (see below).

effects mask substantial variation over time and across different segments of the border, which I turn to next.

Panels (a) and (b) of Figure 5 show point estimates and 95% confidence intervals for the interaction between the border dummy and a full set of year indicators (instead of the post-1999 Chávez's time dummy), using my preferred specification. Panel (a) shows results for FARC Events, while the dependent variable in panel (b) is ELN Events. Looking at panel (a), I find that FARC activity in border municipalities increased by roughly 0.25 SD as soon as Chávez came to power in 1999 (relative to the omitted year - 1993) and remained high until 2003. It decreased somewhat in 2004 but increased again between 2005 and 2007. This pattern is consistent with the anecdotal evidence discussed in section 2.2 on the deterioration and posterior rekindling of the insurgents' relation with the Venezuelan government during those years. Furthermore, the graph shows that the difference in FARC activity between border municipalities and the rest of the country was stable between 1993 and 1998, lending credibility to the parallel trends assumption underlying the empirical strategy.

Regarding ELN, the results in panel (b) indicate that the intensity of the group's activities in border municipalities rose throughout most of the 1990s (which led to the Caldera border policies mentioned in section 2.2). However, there is evidence of a statistically significant increase in ELN activity in these municipalities during the early years of the Chávez administration, but also of a substantial decrease after 2003.<sup>28</sup>

The decline in ELN activity is consistent with FARC's expansion into the group's stronghold in Arauca, which can be seen in panels (c) and (d) of Figure 5. These panels show results from another modified version of equation (1) that includes interactions of the Chávez time indicator with separate border dummies for each Colombian department.<sup>29</sup> Panel (c) shows that FARC acti-

<sup>&</sup>lt;sup>28</sup>If I regress ELN Events on separate interactions of D(Border Venezuela)<sub>i</sub> with the early (1999-2003) and late Chávez periods (2004-2008), the point estimates (standard errors) using my preferred specification are 0.22 (0.10) and -0.20 (0.15), respectively.

<sup>&</sup>lt;sup>29</sup>I join Boyacá with Arauca and Vichada with Guainía due to the small number of municipalities in these departments.

vity increased throughout the border area during the Chávez years, with the exception of Vichada and Guainía in the southernmost section (see Figure 3a). The point estimates for Guajira, Cesar and North Santander are all around 0.25 SD, but the standard error is larger for the latter (p=0.106). The increase in the segment of the border corresponding to Arauca was much larger (0.72 SD)and is matched by an equally large decrease in ELN activity in that segment (-0.83 SD), shown in panel (d). These findings match the information made public by the Colombian government on FARC's bases in Venezuela, which were near Guajira and Cesar but mostly in the Venezuelan state of Apure, near Arauca (Robinson, 2003; El Espectador, 2010a; IISS, 2011a). Panel (d) also shows that ELN events rose 0.3 SD (p=0.013) during the Chávez administration in the group's other traditional stronghold, the segment of the border belonging to North Santander.

Going back to Table 2, I find that the increase in insurgent activity at the border with Venezuela during the Chávez years led to increased violence and to the deterioration of local living conditions. Column 6 shows that the homicide rate in border municipalities increased by almost 20 murders per 100,000 inhabitants, which is a very large increase (35%) relative to the sample mean of 56 per 100,000 inhabitants. In columns 7 and 8, I also observe decreases in the main sources of local tax revenue, the business and property taxes. These reductions are also quite large and represent a more than 30% decrease relative to the sample mean. They reflect the deterioration of local state capacity and the weakening of the local economy as a result of increased insurgent activity.<sup>30</sup> Additional insurgent activity also has negative effects on human capital accumulation.<sup>31</sup> The results in column 9 show that educational enrolment (measured as the percentage of 5-24 year-olds attending an educational institution) decreased in border municipalities after 1999 by 6 percentage points (10% decrease relative to sample mean).

Most of these negative outcomes are also observed in one-from-border mu-

<sup>&</sup>lt;sup>30</sup>On the negative economic effects of violence, see Abadie and Gardeazabal (2003); Besley and Mueller (2012); Camacho and Rodríguez (2013); Besley et al. (2015); Rozo (2016).

<sup>&</sup>lt;sup>31</sup>For more information, see Ichino and Winter-Ebmer (2004); Akresh et al. (2012a,b); León (2012); Herrera (2014).

nicipalities, but the magnitude of the coefficient is always smaller than in border municipalities. This indicates that increased cross-border insurgent activity affected mostly border municipalities, but also had negative spill-over effects on neighboring municipalities.

#### 6.2. Heterogeneity across types of insurgent activity

As discussed in section 3, the availability of a cross-border refuge is likely to enable the insurgents to carry out particularly risky activities that would otherwise be unfeasible. Riskier actions involve many operatives or require a long time to be carried out, leaving the insurgents more exposed to a response by government forces. To explore this hypothesis, I estimate equation (1) using as dependent variable each of the ten disaggregate indicators of FARC activity that have at least one non-zero value every year.

The results, shown in panel A of Table 3, indicate that the largest increases in FARC activity were in town sieges (0.31 SD), terrestrial piracy (0.30 SD), explosive terrorist acts (0.26 SD) and armed contact with government forces (0.23 SD). These results prove that the findings from the previous section are not driven by one specific type of insurgent activity. More importantly, they bring to light the high cost of transnational insurgency in terms of political violence and conflict intensity.

The activities that experienced the largest increases, except perhaps for acts of terrorism, imply substantial risks for the insurgents. For example, town sieges require many operatives to surround the urban centre of a municipality and exert sustained military pressure on local police and other government forces stationed there. The example from the FARC incursion in Tibú in section 3 illustrates how increased access to Venezuela facilitated these actions, as FARC operatives could attack when it was most convenient and withdraw when military reinforcements arrived. Similarly, the availability of a crossborder refuge in which to store loot made it more profitable to carry out acts of land piracy.

A more nuanced picture emerges if we look separately at the first five years

of the Chávez administration (1999-2003) and at the remaining years for which data is available (2004-2008). This is a useful exercise because the latter period could also reflect an endogenous response in counterinsurgency to the increase in insurgent activity in the border area (Berman et al., 2011). The results in panel B show that terrestrial piracy and town sieges (columns 6 and 9), the two risky activities just discussed, were the ones that grew the most in border municipalities during the first years of the Chávez administration. Column 8 shows that ambushes also grew (another risky activity that requires detailed preparations and many operatives), as well as murder and explosive terrorist acts (columns 1 and 2).

In the later years, we observe a substitution from terrestrial piracy, ambushes and town sieges to armed contact with government forces (column 10). This result suggests that security improved as the government's counterinsurgency efforts adjusted to the heightened insurgent threat in the border area. The large increase in explosive terrorist acts in the later period (column 2) provides additional evidence that improved counterinsurgency forced FARC to resort to more irregular tactics.

#### 6.3. Robustness Checks

The results presented above provide evidence of a large and costly increase in FARC activity at the border with Venezuela after Chávez became president. In this section I provide various tests of the robustness of this finding. Unless otherwise stated, for these tests I employ my preferred specification with municipality and region-year fixed effects, plus additional controls (column 3 of Table 2).

I first test whether the main result is robust to the use of the DV dataset on the Colombian conflict. This is a relevant test as the correlation between the CEDE and DV datasets is not very high.<sup>32</sup> Column 1 in Table 4 shows

<sup>&</sup>lt;sup>32</sup>The correlation between "FARC Events" from CEDE and "Guerrilla Events" from DV is only 0.33. Of course, "FARC Events" is constructed adding 19 activity indicators while "Guerrilla Events" is the sum of only three, but even for a single and comparable indicator, such as the number of political kidnappings, the correlation is only 0.55.

estimates of equation (1) with Guerrilla events, constructed using the DV dataset, as the dependent variable. Although data from this source is available for a smaller number of municipalities and for a shorter time period (1993-2004), the standardized estimates are remarkably similar to those obtained using the CEDE conflict dataset and they are equally precise.

The DV data also allows me to benchmark the increase in guerrilla activity at the border with Venezuela caused by Chávez against the effect of commodity price shocks reported in Dube and Vargas (2013). If I re-estimate the model using the number of guerrilla attacks as the dependent variable (without normalizing for population or standardizing), the point estimate for D(Chávez)<sub>t</sub> x D(Border Venezuela)<sub>i</sub> is 0.96 (p=0.012). This effect is one order of magnitude larger than the increase in guerrilla attacks that Dube and Vargas (2013, p. 1403) find for the average coffee-growing municipality following a 50 % drop in coffee prices.

I next address the concern that disproportionately high growth in coca cultivation in the border region may be confounding the results. This is a valid concern because coca cultivation expanded dramatically in Colombia during the 1990s and Venezuela became an increasingly important point in drugtrafficking routes (Angrist and Kugler, 2008; Avila, 2012). For this purpose, I expand the set of controls by including the interaction between year dummies and an indicator for presence of coca crops in 2000, which is the earliest date for which municipality-level data is available. These are potentially "bad controls", since the location of coca crops in 2000 could be affected by FARC's increased access to Venezuela after 1999 (Angrist and Pischke, 2008). However, the estimates in column 2 of Table 4 are very robust to their inclusion.

One concern with the findings above is that they could be capturing a disproportionate increase in insurgent activity in municipalities close to any of the country's borders. This could be the case if the borders are endogenously located in areas that are well suited to insurgent activity and the guerrillas retreated to these advantageous areas when the conflict intensified around the turn of the century.<sup>33</sup> For example, the evidence from the seized laptop

<sup>&</sup>lt;sup>33</sup>On the endogenous shape of nations, see Alesina and Spolaore (2003). See Fearon and

of FARC's second-in-command, "Raúl Reyes", indicates that he moved into Ecuador in 2003 (IISS, 2011b). Regression results, displayed in Figure 6, confirm that FARC activity at the border with Ecuador increased after 2003.<sup>34</sup> Hence, it is important to examine whether the growth in FARC activity near Venezuela was part of a more general trend in the geography of insurgent activity.

In column 3 of Table 4, I include the interaction between year dummies and the distance to the nearest of Colombia's borders as additional controls. The results are robust to this modification and, thus, not driven by heterogeneity in guerrilla growth by proximity to a border. A more stringent test is presented in column 4, which uses a reduced sample of municipalities located at any of Colombia's borders and their contiguous non-border neighbors. I find that the results remain unchanged if I only use Colombia's other borders as a control group for the border with Venezuela.

The estimates in column 5 are obtained when I include the interaction between year dummies and the value of the dependent variable in 1998 as additional controls. The results are unaffected by this modification, which implies that the increase in FARC activity after 1999 is not due to border municipalities having different levels of FARC activity before Chávez (i.e. FARC growing more where it was originally stronger).

The results are even robust to the inclusion of a quadratic trend specific to municipalities near the border with Venezuela. This is a particularly stringent test, as the trend is likely to capture a large portion of the variation in FARC activity near the border over time. Nevertheless, the results in column 6 still show a 0.2 SD increase in FARC activity in border municipalities when Chávez was in power (p=0.091).

Column 7 shows results from a negative binomial model with the count of FARC Events (without dividing by population or standardizing) as dependent variable. The model, which includes municipality and year fixed effects, is

Laitin (2003) and Nunn and Puga (2012) on terrain ruggedness and exposure to violence.

<sup>&</sup>lt;sup>34</sup>Figure 6 also shows that FARC Events decreased after Rafael Correa came to power in 2007, which is consistent with the qualitative evidence on FARC's inability to obtain protection from the new Ecuadorian government (IISS, 2011a).

ideal for count data with overdispersion.<sup>35</sup> The results once more point to a statistically significant increase in FARC activity at the border with Venezuela when Chávez was in power.

FARC activity at the border could have also increased due to variation in local economic conditions during the Chávez years.<sup>36</sup> However, column 8 shows that the results are very robust to the inclusion of the interaction between Venezuela's GDP growth rate and the border and one-from-border dummies as additional controls in equation (1). Thus, it is not economic dependence on Venezuela what is leading to greater FARC activity at the border. In column 9, I include the interaction between the price of oil and the indicators for municipalities near Venezuela as additional controls. This regression is motivated by the high dependence of the Venezuelan economy on oil, the price of which rose dramatically during the Chávez years. It is also motivated by the fact that Colombia's main oil-producing region is in the eastern plains, relatively near the border. The results remain unchanged.<sup>37</sup>

To further explore the possibility of local economic shocks, in column 10 I include log real GDP of the department to which the municipality belongs as an additional control in equation (1). This is the most disaggregate level at which GDP is available in Colombia and the data is available up to 2005. Again, the results are quite robust, suggesting that it is not departmental variation in economic conditions what is driving the results.

Finally, I check the robustness of the results to the inclusion of controls for local political conditions. This way, I address the concern that the border region may have tilted politically to the left at the same time as Venezuela did. This is a particularly relevant concern given Chávez's observed willingness to export his political agenda throughout Latin America (Clem and Maingot,

<sup>&</sup>lt;sup>35</sup>The variance of the count of FARC Events is more than one order of magnitude greater than the mean. However, the results are very similar if I use a fixed-effects Poisson model or a zero-inflated negative binomial model with year fixed effects (not reported).

<sup>&</sup>lt;sup>36</sup>On the effect of economic conditions on conflict onset and intensity, see Collier and Hoeffler (2004); Miguel et al. (2004); Dube and Vargas (2013).

<sup>&</sup>lt;sup>37</sup>It is likely that the restrictions on bilateral trade imposed by Chávez had a negative effect on the border economy, but these only began in 2009, after the end of the sample period (BBC News, 2009).

2011). Column 11 shows that the results on FARC Events are robust to the inclusion as controls of the vote share for the winning candidate and the main left-wing party in the last presidential election. I also include separate dummies for the party in control of the local government (Liberal or Conservative).

Figure 7 provides additional evidence on the stability of politicial conditions in the border region over time. It shows point estimates and 95% confidence intervals for the interaction between the border dummy and indicators for election years in a modified version of (1). The dependent variable is specified in the caption. Panels (a) and (b) look at presidential elections, which take place every four years (omitted year is 1986), while panels (c) and (d) look at mayoral elections, which were more frequent during the sample period (term length changed in 1994 and 2003 and the omitted year is 1988).

The results in panel (a) indicate that there was no systematic change in the vote share of the Liberal party (only party to present its own candidate in all presidential elections) in border municipalities when Chávez was in power. Panel (b) shows a similar result for the main left-wing party taking part in the presidential race. The main left-wing party in the 1986 presidential election (omitted year) was "Unión Patriótica" (UP), a party created by FARC and other left-wing groups. Thus, the results in panel (b) indicate that there was no significant change in popular support for left-wing parties relative to the baseline support revealed for FARC's UP party in the 1986 presidential elections. Panel (c) shows that there was also no significant change in the vote share for the two traditional parties (Liberal and Conservative) in the elections for municipal mayor after Chávez came to power in 1999. Panel (d) looks at the competitiveness of elections, using as dependent variable the winner's vote share in the mayoral election. The graph shows that the difference in the winning mayor's vote share between border municipalities and the rest of the country was fairly stable over time.

#### 6.4. Activity by AUC and the Colombian military

In this section I provide evidence against two possible explanations, other than increased FARC access to Venezuela during the Chávez administration, for the previous findings. The first one is the expansion of the paramilitary organization AUC after 1997. The second one is the large contemporary disbursement of US military aid to Colombian and the consequent strengthening of the Colombian military.

As discussed in section 2.1, several atomized paramilitary groups combined to form AUC around 1997. This move was driven to a large extent by FARC's military success at the time, making the defeat of the leftist guerrillas the paramilitaries' main objective. AUC's intense counterinsurgency campaign roughly coincided with the start of the Chávez administration (see Figure 2) and could potentially explain the previous results.

Column 1 of Table 5 shows estimates of equation (1) using AUC Events as the dependent variable. The results indicate that AUC activity decreased in border municipalities after 1999 but increased in one-from-border municipalities, which is consistent with FARC having a military advantage at the border.<sup>38</sup> However, these estimates are imprecise and statistically insignificant. Column 2 further shows that the results on FARC are very robust to controlling for AUC Events, suggesting that the paramilitary expansion is not the mechanism underlying the increase in guerrilla activity at the border. Column 3 shows that the results are also robust to allowing for heterogeneous yearly variation in FARC Events in municipalities that had paramilitary presence (non-zero AUC Events) at any point in the sample period.

Another alternative explanation for the increase in FARC activity at the border with Venezuela after 1999 has to do with improved counterinsurgency around the same time. The U.S. started disbursing the "Plan Colombia" military aid package in 2000. Better funding and the breakdown of peace talks in 2002 led to an agressive military campaign against FARC during the Uribe administration (see Figure 2).

 $<sup>^{38}{\</sup>rm The}$  results are unchanged if I set 2003 as the final year of the sample period to account for the demobilization of AUC.

The results in column 4 of Table 5 provide evidence of a small and imprecisely estimated increase of 0.10 SD in military activity near the border with Venezuela during the Chávez years. This increase is consistent with counterinsurgency efforts following the trail of insurgent activity and crowding out paramilitary activity. More importantly, column 5 shows that the result on FARC Events is robust to controlling for Armed Forces Events, despite the strong correlation between the two variables. Column 6 shows that the results are robust to the inclusion as additional controls of interactions between yearly nationwide FARC losses (captures + demobilizations) and the border and one-from-border dummies. This means that the results are not driven by FARC retreating to the border with Venezuela in years when they were suffering higher losses. Column 7 shows that the results are robust to the inclusion of a similar set of interactions with the yearly amount of U.S. aid instead. The magnitude of the increase in FARC Events in border municipalities drops to 0.2 SD, but this is understandable given that the correlation between U.S. aid and the Chávez time dummy is 0.83.

## 7. Concluding remarks

This paper documents a large and robust increase in FARC activity at the border with Venezuela after Hugo Chávez became president of the latter in 1999. Such an increase is consistent with FARC enjoying increased access to Venezuela during the Chávez administration, but being unable to exploit it beyond the immediate vicinity of the border. I show that the additional guerrilla activity had a high human and material cost. It led to more armed clashes, acts of terrorism and town sieges, as well as to higher homicide rates and lower local tax revenue and educational enrolment rates.

These quantitative findings are consistent with a large body of qualitative evidence in support of the hypothesis that the Chávez administration actively collaborated with Colombian insurgent groups. If this hypothesis is true, the local effects reported for municipalities near Venezuela probably provide a lower bound on the effects of foreign support, as financial assistance and the provision of weapons would have also stimulated guerrilla activity in control municipalities far from the border. However, it is likely that the findings in this paper also reflect a deterioration in border security during the Chávez years (Boraz, 2007; Avila, 2012). Incidents such as the murder of seven Venezuelans by FARC in 2004 certainly suggest that not all of the group's activities inside Venezuela were coordinated with the government.

Regardless of the degree of active collaboration between the government of Venezuela and the Colombian rebels, this paper shows that access to foreign territory allows insurgent groups to expand their operations significantly and leads to large increases in conflict intensity. In consequence, domestic counterinsurgency campaigns may prove unsuccessful if they lack support from neighboring countries or if state capacity in border areas is too weak. But as long as insurgent presence leaves a geographical trail, forensic strategies such as the one employed in this paper can prove useful for detecting and further studying transnational insurgent activities.

#### Acknowledgements

Previous versions of this paper circulated under various titles. Most of the work for this paper was carried out at the London School of Economics. I would like to thank Gerard Padró i Miquel for his support and advice throughout this project, as well as Munir Squires, Miguel Espinosa, Pieter Serneels, Fernando Galano and five anonymous referees for their detailed comments and suggestions. I am also grateful to seminar participants at LSE, Universidad de los Andes, the University of East Anglia's ID PhD conference (2013) and the 9th annual HiCN Workshop (2013). All remaining errors are mine.

# References

- Abadie, A. and Gardeazabal, J. (2003). The Economic Costs of Conflict: A Case Study of the Basque Country. American Economic Review, 93(1):113– 132.
- Acemoglu, D., Robinson, J., and Santos, R. (2013). The Monopoly of Violence: Evidence from Colombia. Journal of the European Economic Association, 11(S1):5–44.

- Acevedo, K. and Bornacelly, I. (2014). Panel Municipal del CEDE. Documento CEDE 2014-26.
- Akresh, R., Bhalotra, S., Leone, M., and Osili, U. O. (2012a). War and Stature: Growing Up during the Nigerian Civil War. American Economic Review, 102(3):273–77.
- Akresh, R., Lucchetti, L., and Thirumurthy, H. (2012b). Wars and Child Health: Evidence from the Eritrean–Ethiopian Conflict. *Journal of Development Economics*, 99(2):330 – 340.
- Albornoz, F. and Hauk, E. (2014). Civil War and U.S. Foreign Influence. Journal of Development Economics, 110:64–78.
- Alesina, A. and Spolaore, E. (2003). The Size of Nations. MIT Press, Cambridge, MA.
- Angrist, J. D. and Kugler, A. D. (2008). Rural Windfall or a New Resource Curse? Coca, Income and Civil Conflict in Colombia. *Review of Economics* and Statistics, 90(2):191–215.
- Angrist, J. D. and Pischke, J.-S. (2008). Mostly Harmless Econometrics: An Empiricist's Companion. Princeton University Press, Princeton, NJ.
- Avila, A. F., editor (2012). La Frontera Caliente entre Colombia y Venezuela. Arcoiris / Debate, Bogotá, Colombia.
- BBC News (2004). Colombia Captures Rebel 'Envoy'. December 16.
- BBC News (2005a). Colombia Makes 'Record Arms Haul'. May 30.
- BBC News (2005b). Venezuela Cuts Ties with Colombia. January 14.
- BBC News (2007). Chávez Meets Colombia Farc Rebels. November 08.
- BBC News (2008a). Colombia backs Chavez Farc call. June 10.
- BBC News (2008b). Colombia Neighbours Deploy Troops. March 03.
- BBC News (2008c). Colombian Hostages Freed by Farc. February 28.
- BBC News (2008d). Top Farc Leader Killed by Troops. March 01.
- BBC News (2009). Alarm at US-Colombia Troop Plans. August 10.

- BBC News (2010). Colombia Takes Venezuela Rebel Accusation to OAS. July 18.
- Becker, G. S. (1968). Crime and Punishment: An Economic Approach. *Journal* of Political Economy, 76(2):169–217.
- Berger, D., Corvalan, A., Easterly, W., and Satyanath, S. (2013a). Do Superpower Interventions Have Short and Long Term Consequences for Democracy? *Journal of Comparative Economics*, 41(1):22–34.
- Berger, D., Easterly, W., Nunn, N., and Satyanath, S. (2013b). Commercial Imperialism? Political Influence and Trade during the Cold War. American Economic Review, 103(2):863–896.
- Berman, E., Shapiro, J. N., and Felter, J. H. (2011). Can Hearts and Minds Be Bought? The Economics of Counterinsurgency in Iraq. *Journal of Political Economy*, 119(4):766–819.
- Besley, T., Fetzer, T., and Mueller, H. (2015). The Welfare Cost of Lawlessness: Evidence from Somali Piracy. *Journal of the European Economic Association*, 13(2):203–239.
- Besley, T. and Mueller, H. (2012). Estimating the Peace Dividend: The Impact of Violence on House Prices in Northern Ireland. American Economic Review, 102(2):810–33.
- Betancourt, I. (2011). Even Silence has an End. Virago Press, London, UK.
- Boraz, S. (2007). Case study: the Colombia-Venezuela border. In Ungoverned Territories: Understanding and Reducing Terrorism Risks. RAND Corporation, Santa Monica, CA.
- Bueno de Mesquita, E. (2013). Rebel Tactics. *Journal of Political Economy*, 121(2):323–357.
- Buhaug, H. and Gates, S. (2002). The Geography of Civil War. Journal of Peace Research, 39(4):417–433.
- Buhaug, H. and Rod, J. K. (2006). Local Determinants of African Civil Wars: 1970-2001. Political Geography, 25(1):315–335.
- Byman, D., Chalk, P., Hoffman, B., Rosenau, W., and Brannan, D. (2001). Trends in Outside Support for Insurgent Movements. RAND Corporation, Santa Monica, CA.

- Camacho, A. and Rodríguez, C. (2013). Firm Exit and Armed Conflict in Colombia. Journal of Conflict Resolution, 57(1):89–116.
- Cameron, C., Gelbach, J., and Miller, D. (2011). Robust Inference with Multiway Clustering. Journal of Business and Economic Statistics, 29(2):238–249.
- Caselli, F., Morelli, M., and Rohner, D. (2015). The Geography of Interstate Resource Wars. The Quarterly Journal of Economics, 130(1):267–315.
- Clem, R. and Maingot, A., editors (2011). Venezuela's petro-diplomacy: Hugo Chávez's Foreign Policy. University Press of Florida, Gainesville, FL.
- Collier, P. and Hoeffler, A. (2004). Greed and Grievance in Civil War. Oxford Economic Papers, 56(4):563–595.
- Corrales, J. (2009). Using Social Power to Balance Soft Power: Venezuela's Foreign Policy. The Washington Quarterly, 32(4):97–114.
- Corrales, J. (2013). Explaining Chavismo: The Unexpected Alliances of Radical Leftists and the Military in Venezuela Since the Late 1990s. In Hausmann, R. and Rodríguez, F., editors, Venezuela Before Chávez: Anatomy of a Collapse. Penn State Press, University Park, PA.
- Cortés, D., Vargas, J. F., Hincapié, L., and Franco, M. d. R. (2012). Seguridad Democrática, Presencia de la Policía y Conflicto en Colombia. *Desarrollo y Sociedad*, 69:11–32.
- Cragin, K. and Hoffman, B. (2003). Arms Trafficking and Colombia. RAND Corporation MR-1468.
- Cunningham, D., Gleditsch, K. S., and Salehyan, I. (2007). Non-State Actors in Civil Wars. Dataset.
- Dell, M. (2015). Trafficking Networks and the Mexican Drug War. American Economic Review, 105(6):1738–1779.
- Doyle, M. W. and Sambanis, N. (2000). International Peacebuilding: A Theoretical and Quantitative Analysis. American Political Science Review, 94(4):779–801.
- Dube, A., Dube, O., and García-Ponce, O. (2013). Cross-Border Spillover: U.S. Gun laws and Violence in Mexico. American Political Science Review, 107(3):397–417.

- Dube, A., Kaplan, E., and Naidu, S. (2011). Coups, Corporations, and Classified Information. Quarterly Journal of Economics, 126(3):1375–1409.
- Dube, O. and Naidu, S. (2015). Bases, Bullets and Ballots: The Effect of US Military Aid on Political Conflict in Colombia. *Journal of Politics*, 77(1):249–267.
- Dube, O. and Vargas, J. F. (2013). Commodity Price Shocks and Civil Conflict: Evidence from Colombia. *Review of Economic Studies*, 80(4):1384–1421.
- El Espectador (2010a). Farc tienen 1500 hombres en 28 campamentos en Venezuela. July 17.
- El Espectador (2010b). Los Tentáculos de las FARC en Venezuela. May 18.
- El Espectador (2014). Eln y el Narcotráfico: Una Relación Peligrosa. July 07.
- El Tiempo (1995a). Caldera Ordena Militarizar Frontera. March 16.
- El Tiempo (1995b). Venezuela Cruzaría Frontera en Legítima Defensa: Caldera. March 19.
- El Tiempo (1996). Venezuela Insiste en la Persecución en Caliente. January 07.
- El Tiempo (1997). Venezuela Refuerza Seguridad Fronteriza. February 20.
- El Tiempo (1998). Venezuela Autorizó Persecución en Caliente. May 03.
- El Tiempo (1999). Nuevos Detalles de Relación Chávez-FARC. August 27.
- El Tiempo (2000a). Chávez Promueve Entrega de Armas a Guerrilla Colombiana. June 17.
- El Tiempo (2000b). Venezuela Rechazó Denuncia de Tapias. July 12.
- El Tiempo (2002). FARC: Base en Venezuela. March 22.
- El Tiempo (2004a). Campamento del ELN en Mérida. June 23.
- El Tiempo (2004b). Muertos en la Frontera Llegan a 14. September 24.
- El Tiempo (2004c). Venezuela Deportó a Ocho Guerrilleros. June 21.
- El Tiempo (2005a). FARC y ELN Combaten en Frontera Venezolana. March 30.

- El Tiempo (2005b). Revive Guerra entre Farc y Eln: 9 muertos. December 16.
- El Tiempo (2008). Calificativo de Terroristas No se les Levantará a las Farc ni al Eln: Gobierno de Colombia. January 10.
- El Tiempo (2010). Chávez Cuestiona la Lucha Armada. July 24.
- El Tiempo (2012). Miembros de las Farc y el Eln se han Disminuido, Asegura Gobierno. December 27.
- El Universal (2002). El Gobierno Ofreció Apoyo a la Guerrilla Colombiana. January 30.
- Fearon, J. and Laitin, D. (2003). Ethnicity, Insurgency, and Civil War. American Political Science Review, 97(1):75–90.
- Fergusson, L., Robinson, J., Torvik, R., and Vargas, J. F. (2014). The Need for Enemies. *Economic Journal*, 126:1018–1054.
- Gleditsch, K. S. (2007). Transnational Dimensions of Civil War. Journal of Peace Research, 44(3):293–309.
- Gleditsch, K. S., Salehyan, I., and Schultz, K. (2008). Fighting At Home, Fighting Abroad: How Civil Wars Lead to International Disputes. *Journal* of Conflict Resolution, 52(4):479–506.
- Gonsalves, M., Stansell, K., and Howes, T. (2009). Out of Captivity: Surviving 1,967 Days in the Colombian Jungle. HarperCollins, New York, NY.
- Grier, K. and Maynard, N. (2016). The Economic Consequences of Hugo Chavez: A Synthetic Control Analysis. Journal of Economic Behavior & Organization, 125:1–21.
- Hegre, H. and Sambanis, N. (2006). Sensitivity Analysis of Empirical Results on Civil War Onset. Journal of Conflict Resolution, 50(4):508–535.
- Herrera, J. C. (2014). Petróleo y Desarrollo: Efectos en la Acumulación y Destrucción de Capital Humano de los Municipios de Colombia. Documento CEDE 2014-10.
- Hsieh, C.-T., Miguel, E., Ortega, D., and Rodríguez, F. (2011). The Price of Political Opposition: Evidence from Venezuela's Maisanta. American Economic Journal: Applied Economics, 3(2):196–214.

- ICG (2004). Colombia's Borders: The Weak Link in Uribe's Security Policy. International Crisis Group. Latin America Report No. 9.
- Ichino, A. and Winter-Ebmer, R. (2004). The Long-Run Educational Cost of World War II. Journal of Labor Economics, 22(1):57–87.
- IISS (2011a). The FARC files: Venezuela, Ecuador and the secret archive of 'Raúl Reyes'. The International Institute for Strategic Studies.
- IISS (2011b). The FARC files: Venezuela, Ecuador and the secret archive of 'Raúl Reyes' (Companion CD). The International Institute for Strategic Studies.
- Kalyvas, S. N. (2006). The Logic of Violence in Civil War. Cambridge University Press, New York, NY.
- Kalyvas, S. N. and Balcells, L. (2010). International System and Technologies of Rebellion: How the End of the Cold War Shaped Internal Conflict. *American Political Science Review*, 104(3):415–429.
- León, G. (2012). Civil Conflict and Human Capital Accumulation: The Longterm Effects of Political Violence in Perú. Journal of Human Resources, 47(4):991–1022.
- Lindsay, F. A. (1962). Unconventional Warfare. *Foreign Affairs*, 40(2):264–274.
- Malet, D. (2013). Foreign Fighters: Transnational Identity in Civil Conflicts. Oxford University Press, New York, NY.
- McCoy, J. (1999). Chavez and the End of "Partyarchy" in Venezuela. *Journal* of *Democracy*, 10(3):64–77.
- Michalopoulos, S. and Papaioannou, E. (2016). The Long-Run Effects of the Scramble for Africa. *American Economic Review*, 106(7):1802–1848.
- Miguel, E. and Roland, G. (2011). The Long-Run Impact of Bombing Vietnam. Journal of Development Economics, 96(1):1–15.
- Miguel, E., Satyanath, S., and Sergenti, E. (2004). Economic Shocks and Civil Conflict: An Instrumental Variables Approach. *Journal of Political Economy*, 112(4):725–753.

- Mueller, H. (2016). Growth and Violence: Argument for a Per Capita Measure of Civil War. *Economica*, 83:473–497.
- Nunn, N. and Puga, D. (2012). Ruggedness: The Blessing of Bad Geography in Africa. *Review of Economics and Statistics*, 94(1):20–36.
- Nunn, N. and Qian, N. (2014). U.S. Food Aid and Civil Conflict. American Economic Review, 104(6):1630–1666.
- Ortega, D. and Rodríguez, F. (2008). Freed from Illiteracy? A Closer Look at Venezuela's Misión Robinsón Literacy Campaign. *Economic Development* and Cultural Change, 57(1):1–30.
- Otis, J. (2014). *The FARC and Colombia's Illegal Drug Trade*. Woodrow Wilson International Center for Scholars. Latin American Program.
- Pachón, M. and Sánchez, F. (2014). Base de Datos sobre Resultados Electorales CEDE, 1958-2011. Documento CEDE 2014-19.
- Palacios, M. (2012). Violencia Pública en Colombia, 1958-2010. Fondo de Cultura Económica, Bogotá, Colombia.
- Pizarro, E. (2007). La Insurgencia Armada: Raíces y Perspectivas. In Sánchez, G. and Peñaranda, R., editors, *Pasado y Presente de la Violencia en Colombia*. La Carreta Editores, Bogotá, Colombia.
- Ramírez, S. (2003). Colombia Venezuela: Entre Episodios de Cooperación y Predominio del Conflicto. In Domínguez, J., editor, *Conflictos Territoriales* y Democracia en América Latina. Siglo XXI, Buenos Aires, Argentina.
- Rangel, A. (2000). Parasites and Predators: Guerrillas and the Insurrection Economy of Colombia. *Journal of International Affairs*, 53(2):577–601.
- Regan, P. M. (2002). Third-party Interventions and the Duration of Intrastate Conflicts. Journal of Conflict Resolution, 46(1):55–73.
- Restrepo, J., Spagat, M., and Vargas, J. F. (2004). The Dynamics of the Colombian Civil Conflict: A New Dataset. *Homo Oeconomicus*, 21(2):396– 428.
- Richani, N. (2013). Systems of Violence: The Political Economy of War and Peace in Colombia. State University of New York Press, Albany, NY.

- Robinson, L. (2003). Terror Close to Home: A Volatile Leader Befriends Mideast, Colombia and Cuba. U.S. News and World Report, 135(11).
- Rodríguez, F. (2008). An Empty Revolution: The Unfulfilled Promises of Hugo Chávez. Foreign Affairs, 87(2):49–62.
- Rohner, D., Thoenig, M., and Zilibotti, F. (2013). Seeds of Distrust: Conflict in Uganda. *Journal of Economic Growth*, 18(3):217–252.
- Rosendorff, B. P. and Sandler, T. (2005). The Political Economy of Transnational Terrorism. *Journal of Conflict Resolution*, 49(2):171–182.
- Rozo, S. V. (2016). Is Murder Bad for Business? Evidence from Colombia. Working Paper.
- Salehyan, I. (2007). Transnational Rebels: Neighboring States as Sanctuary for Rebel Groups. World Politics, 59(2):217–242.
- Salehyan, I. (2008a). No Shelter Here: Rebel Sanctuaries and International Conflict. Journal of Politics, 70(1):54–66.
- Salehyan, I. (2008b). The Externalities of Civil Strife: Refugees as a Source of International Conflict. American Journal of Political Science, 52(4):787– 801.
- Salehyan, I. (2009). Rebels Without Borders: Transnational Insurgencies in World Politics. Cornell University Press, Ithaca, NY.
- Salehyan, I. (2010). The Delegation of War to Rebel Organizations. Journal of Conflict Resolution, 54(3):493–515.
- Salehyan, I. and Gleditsch, K. S. (2006). Refugees and the Spread of Civil War. International Organization, 60(2):335–366.
- Salehyan, I., Gleditsch, K. S., and Cunningham, D. (2011). Explaining External Support for Insurgent Groups. *International Organization*, 65(4):709– 744.
- Semana (1995). Caldera Caliente. April 10.
- Semana (2002). Relaciones Peligrosas. February 26.
- Semana (2005). Las Cuentas de las FARC. January 30.

Semana (2009). Los 'Rockets' Venezolanos. July 25.

- Serneels, P. and Verpoorten, M. (2015). The Impact of Armed Conflict on Economic Performance: Evidence from Rwanda. *Journal of Conflict Resolution*, 59(4):555–592.
- The New York Times (2009a). Rebels Obtained Arms Sold to Venezuela, Colombia Says. July 27.
- The New York Times (2009b). Venezuela Still Aids Colombia Rebels, New Material Shows. August 02.
- The Washington Post (2007). Colombia Cancels Chávez Mediation Role. November 22.
- U.S. Army (1990). Foot Marches. Field Manual 21-18.
- Weinstein, J. M. (2007). *Inside Rebellion: The Politics of Insurgent Violence*. Cambridge University Press, New York, NY.
- Wright, A. (2016). Economic Shocks and Rebel Tactics. Working Paper.



Figure 1: FARC activity at the border with Venezuela

Note: The graph shows yearly averages and 95% confidence intervals of the variable "FARC Events" for municipalities at the border with Venezuela and for all others. FARC Events is the sum of 19 activity indicators, divided by 1993 population.

Number of Events by Colombian Armed Forces Number of Events by Illegal Armed Groups Year FARC ELN AUC Armed Forces --

Figure 2: Activity by non-state armed groups and the Colombian military

Note: The graph shows the aggregate number of events per year for each agent. For FARC, ELN and AUC, "Events" is the sum of 19 activity indicators (left axis), while for the Armed Forces it is the sum of 9 indicators (right axis).

Figure 3: Measuring proximity to Venezuela







Figure 4: The geographic distribution of FARC Events

Note: The maps show the number of FARC events per 10,000 inhabitants. Events is the sum of 19 activity indicators. Areas in grey correspond to municipalities with missing data.



Figure 5: Heterogeneity in guerrilla activity in border municipalities

Note: Panels (a) and (b) show point estimates and 95% confidence intervals for the interaction between year dummies and the indicator for border municipalities in a regression with FARC and ELN Events as dependent variable. The regression includes municipality and region-year fixed effects, as well as yearly interactions with fixed municipality characteristics and with the one-from-border dummy(not reported). The omitted year is 1993. The standard errors have been clustered two-way by municipality and department-year. The dependent variable is the standardized sum of 19 indicators of activity by the group in the caption divided by 1993 population. Panels (c) and (d) show results from a regression with identical controls, except that the Chávez time dummy has been interacted with a series of segment-specific "Border" and "One-from-border" dummies. All regressions use 17,338 observations from 1,099 municipalities.



Figure 6: FARC activity at the border with Ecuador

Note: The graph shows point estimates and 95% confidence intervals for the interaction between year dummies and an indicator for municipalities at the border with Ecuador in a regression with FARC events as dependent variable. The omitted year is 1993. The regression includes municipality and region-year fixed effects, as well as yearly interactions with fixed municipality characteristics and a "one-from-border-with-Ecuador" dummy. The standard errors have been clustered two-way by municipality and department-year. The dependent variable is the standardized sum of 19 indicators of activity by FARC divided by 1993 population.



Figure 7: Electoral results in border municipalities

Note: The graphs show point estimates and 95% confidence intervals for the interaction between dummy variables for each presidential (mayoral) election year and the border dummy. The omitted year is 1986 for presidential elections, 1988 for mayoral ones. The regression includes municipality and region-year (election) fixed effects. It also includes a full set of election year interactions with fixed municipality characteristics and with a dummy for one-from-border municipalities. The standard errors have been clustered two-way by municipality and department-year. The dependent variable in panel (a) is the share of votes (0-1) for the candidate of the Liberal party in that year's presidential election. In panel (b) it is the share of votes (0-1) for the candidate of the main left-wing party in that year's presidential election: UP (1986), M19 (1990-1998), PDI (2002), PDA (2006). In panel (c) it is the share of votes (0-1) for the two traditional parties (Liberal and Conservative) in that year's mayoral election. In panel (d) it is the share of votes (0-1) for the winning party in that year's mayoral election.

Variable		E.	ull sample	~	~	Border from-bord	or One- er $(N=77)$	Rest of country (	of the N=1,022)	(01) p-value
	Mean	Std. Deviation	Min.	Max.	N	Mean	Std. Deviation	Mean	Std. Deviation	,
		PANEL A:	Panel-leve	I Variables						
FARC Events <sup>1</sup>	1.11	3.88	0	149.37	17338					
ELN Events <sup>1</sup>	0.33	1.55	0	89.15	17338					
AUC Events <sup>1</sup>	0.14	0.75	0	27.45	17319					
Guerrilla Events <sup>1</sup> $(DV)$	0.37	1.12	0	39.15	11736					
Armed Forces Events <sup>1</sup>	2.21	7.67	0	340.61	17328					
Murder Rate (per 100,000 inh.)	55.93 e.e. 9e	83.72 0.70	0061	2724.71	17338 205 <i>6</i>					
9-24 year-outs avenuing curcational evaluation (70) Promerty fay revenue (Millions of 1998 COP) <sup>1</sup>	100.17	3.12 146 79	00.1	3079.55	17338					
Business tax revenue (Millions of 1998 $COP)^1$	67.96	228.7	0	5057.36	17338					
PANEL B: Munic	cipality-lev	el Variables	proximity	to borders	with Vene	zuela and E	cuador)			
Border with Venezuela (dummv)	0.03	0.18	0	1	1099					
One-from-border with Venezuela (dummy)	0.04	0.19	0	1	1099					
Distance to border with Venezuela (Km)	360.07	232.36	0.24	938.79	1099					
Border with Ecuador (dummy)	0.009	0.09	0	1	1099					
One-from-border with Ecuador (dummy)	0.02	0.14	0	1	1099					
PA PA	ANEL C: M	[unicipality-]	evel Variab	les (other c	haracteris	tics)				
$Area (Km^2)$	947.35	3090.58	15	65674	1099	3516.23	8405.11	753.80	2116.18	0.00
Altitude (Metres above sea level)	1148.07	906.15	2	3087	1099	628.70	739.08	1187.20	905.81	0.00
Distance to nearest market (Km)	123.46	97.61	0	926.47	1099	182.27	119.40	119.03	94.36	0.00
Unmet Basic Needs index in 1993	54.17	19.78	9.15	100	1099	62.60	16.99	53.54	19.83	0.00
Share of rural population in 1993	0.64	0.23	0	1	1099	0.56	0.25	0.64	0.23	0.00
Koyalties in 1998 (million $COP$ ) <sup>1</sup>	91.62 1976 14	584.04 707 00	0 0	10340.44	1099	214.58	638.64 400 70	82.36 1997 21	579.00 71.6.00	0.06
Transfers in 1998 (million $COP$ ) <sup>2</sup>	12/0.14	705.90		7800.84	1000	06.1211	499.50	1287.31	718.00	0.00
D-1	0.0	07.0			ACU1	0.14	0.00	00.0	0.2.0	10.0
broken-up municipanty (auniny) Notary office in 1006 (dummy)	00	070			1000	0.40	0.40	00 0.49	0.49	0.00 0.70
Agricultural hank office in 1996 (dummy)	10.0	0.29	~ _	. –	1099	0.84	0.37	26 U	0.28	0.03
Tax collection office in 1996 (dummy)	0.44	0.50	0		1099	0.39	0.49	0.44	0.50	0.37
Health center or hospital in 1996 (dummy)	0.74	0.44	0	Ц	1099	0.79	0.41	0.74	0.44	0.30
FARC demilitarized zone and neighbors (dummy)	0.02	0.14	0	1	1099	0.01	0.11	0.02	0.14	0.65

**Table 1:** Summary statistics

	(1) FARC	(2) FARC	(3) FARC	(4) FARC	(5) ELN	(6) Murder	(7) Business	(8) Pronerty	(9) 5-24 vo
VARIABLES	Events	Events	Events	Events	Events	Rate	Tax	Tax	in school
$\mathbf{D}(Chávez)_t \ge \mathbf{D}(Border\ Venezuela)_i$	$0.240^{**}$	$0.198^{*}$	0.318*** [0.0050]		0.0115 [0.105]	$19.58^{*}$	$-25.61^{*}$	-33.13*** [0.130]	-5.97** [3 005]
$\mathrm{D}(\mathrm{Chávez})_t \ge \mathrm{D}(\mathrm{One-from-border}$ Venezuela)_i	0.227	0.134 0.134 0.150]	0.117 0.117 0.117		-0.0479 -0.0479 -0.115	$\begin{bmatrix} 11.20 \\ 13.32^{*} \\ 770 \end{bmatrix}$	[14.14] -2.640 [14 09]	[9.120] -20.40*** [E.050]	$[2.636^{*}]$
$\mathbf{D}(\mathbf{C}\mathbf{h}\mathbf{\hat{a}}\mathbf{vez})_t \ge \mathbf{D}(\mathbf{D}\mathbf{istance \ Venezuela}{\leq}10 \ \mathrm{Km})_i$	[061.0]	[net n]	[201.0]	$0.398^{**}$	611.0	[111]	[14:00]	0.302	[1.40 <i>l</i> ]
$\mathrm{D}(\mathrm{Chávez})_t \ge \mathrm{D}(10~\mathrm{Km}{<}\mathrm{Distance~Venezuela}{\leq}20~\mathrm{Km})_i$				0.201* 0.201*					
$\mathrm{D}(\mathrm{Chávez})_t \ge \mathrm{D}(20~\mathrm{Km}{<}\mathrm{Distance~Venezuela}{\leq}30~\mathrm{Km})_i$				0.186* 0.186*					
$\mathbf{D}(\mathbf{Chávez})_t \ge \mathbf{D}(30~\mathrm{Km}{<}\mathrm{Distance~Venezuela}{\leq}100~\mathrm{Km})_i$				$\begin{bmatrix} 0.0931\\ 0.0252\\ 0.0663\end{bmatrix}$					
Observations	17,338	17,338	17,338	17,338	17,338	17,338	17,338	17,338	2,056
Number of municipalities	1,099	1,099	1,099	1,099	1,099	1,099	1,099	1,099	1,028
Year FE	$\mathbf{YES}$	YES	NO	ON	NO	NO	NO	NO	NO
Controls	NO	YES	YES	YES	YES	YES	YES	$\mathbf{YES}$	YES
Region-Year FE	NO	NO	$\mathbf{YES}$	YES	YES	YES	$\mathbf{YES}$	$\mathbf{YES}$	YES
Notes: Dependent variable in the header: FARC (ELN) Ev by 1993 nonulation (ner 10,000 inh.): Murder rate (colum	vents in co nn 6) ner 1	lumns 1-5 100.000 in	is the stan	dardized su	um of 19 in ertv Tax r	dicators of	activity by humns 7-8)	- FARC (ELI are in millio	V) divided us of 1998
Colombian pesos per 10,000 inh.; in column 9, the depende	ent variable	e is the per	centage of	population	aged 5-24	attending	an educatio	nal institutio	n. Sample
period is 1993-2008, except for column 9 which only uses	s data from	1 1993 and	l 2005 (pop	ulation cen	Il All I	egressions	include mu	nicipality fix	ed effects.
Additional controls (columns 2-9) are a full set of year in Standard errors clustered by municipality and department	nteractions t-year in b	s with fixe rackets. *'	d municipa ** p<0.01,	llity charac ** p<0.05,	teristics (s * p<0.1	see panel C	of Table 1	l and text fo	r details).

**Table 2:** The Chávez administration and insurgent activity near the border with Venezuela

			ogonon9							ncra
	(1) Murder	(2) Fxnlosive	(3) Other	(4) Assaults to	(5) Kidnanning	(6) Terrestrial	(7) Illeøal	(8) Ambiishes	(9) Town	(10) Armed
VARIABLES	of	terrorist	terrorist	private	outdourse	piracy	checkpoints		sieges	contact
	civilians	acts	acts	property	civilians	acts				
		PANEL /	A: All Chá	vez years (1)	<u> 999-2008)</u>					
$D(Chávez)_t \ge D(Border Venezuela)_i$	0.0834	$0.263^{**}$	0.0761	$0.128^{*}$	$0.131^{*}$	$0.296^{***}$	$0.142^{**}$	0.0591	$0.309^{**}$	$0.228^{***}$
	[0.0639]	[0.108]	[0.0572]	[0.0747]	[0.0725]	[0.110]	[0.0691]	[0.0375]	[0.146]	[0.0852]
$\mathbf{D}(\mathbf{Chávez})_t \ge \mathbf{D}(\mathbf{One-from-border}\ \mathbf{Venezuela})_i$	0.000649 [0.115]	-0.0148 $[0.0998]$	0.0548 [0.0798]	0.135 [0.0855]	0.102 $[0.0734]$	0.142 [0.113]	0.0792 $[0.0967]$	$0.108^{*}$ $[0.0629]$	0.0892 [0.134]	0.0866 [0.124]
	PANEL	B: Early (19	99-2003) aı	nd late (200	4-2008) Cháv	ez years				
$D(1999-2003)_t \ge D(Border Venezuela)_i$	$0.163^{**}$	$0.148^{*}$	0.0927	0.111	0.104	$0.394^{**}$	0.103	$0.132^{***}$	$0.569^{**}$	0.107
~	[0.0721]	[0.0828]	[0.0694]	[0.0824]	[0.0978]	[0.155]	[0.0891]	[0.0494]	[0.272]	[0.0654]
$D(2004-2008)_t \ge D(Border Venezuela)_i$	0.00415	$0.378^{**}$	0.0596	0.145	$0.158^{*}$	0.198	$0.182^{***}$	-0.0137	0.0494	$0.349^{***}$
	[0.0814]	[0.182]	[0.0622]	[0.110]	[0.0856]	[0.126]	[0.0685]	[0.0671]	[0.0922]	[0.135]
$D(1999-2003)_t \ge D(One-from-border Venezuela)_i$	-0.0440	-0.0784***	0.0695	0.0593	-0.0257	0.0463	-0.0478	0.0466	0.133	-0.0874
	[0.0908]	[0.0290]	[0.107]	[0.0856]	[0.0779]	[0.105]	[0.0751]	[0.0514]	[0.169]	[0.0631]
$D(2004-2008)_t \ge D(One-from-border Venezuela)_i$	0.0453	0.0488	0.0400	0.211	$0.231^{**}$	0.238	0.206	$0.170^{*}$	0.0456	0.261
	[0.159]	[0.187]	[0.0842]	[0.147]	[0.108]	[0.189]	[0.142]	[0.0990]	[0.132]	[0.218]
Observations	17,338	17,338	17,338	17,338	17,338	17,338	17,338	17,338	17,338	17,338
Number of municipalities	1,099	1,099	1,099	1,099	1,099	1,099	1,099	1,099	1,099	1,099
Notes: The dependent variable is the standardized region-war fixed effects They also include a full	l sum of the e	events in the laterations w	neader carri ith fived m	ed out by FA micinality ch	.RC, divided b aracteristics (	y 1993 popula see nanel C o	ttion. All regre f Table and to	essions includ	e municipa.	lity and
clustered by municipality and department-year in	brackets. **	* p<0.01, **	p<0.05, * 1	p<0.1.		o o parror o o	T TODT 1		). Duamman	e10110 1

oss FARC activities near the horder with Venezuela acre oeneitv **Table 3:** The Chávez administration and hete

VARIABLES	(1) Guerrilla Events	(2) FARC Events	(3) FARC Events	(4) FARC Events	(5) FARC Events	(6) FARC Events	(7) FARC Events (count)	(8) FARC Events	(9) FARC Events	(10) FARC Events	(11) FARC Events
$D(Chávez)_t \ge D(Border Venezuela)_i$	$0.396^{**}$ [0.140]	$0.266^{***}$ $[0.0897]$	$0.357^{***}$ [0.0978]	$0.280^{**}$ [0.111]	$0.298^{***}$ $[0.0941]$	$0.205^{*}$ $[0.121]$	$0.412^{**}$ [0.112]	$0.319^{***}$ [0.0923]	$0.325^{***}$ $[0.107]$	$0.302^{***}$ $[0.0904]$	$0.326^{***}$ $[0.0924]$
$D(Chávez)_i \ge D(One-from-border Venezuela)_i$	0.0607 $[0.0950]$	0.0498 $[0.128]$	0.149 $[0.134]$	0.111 [0.116]	0.123 $[0.130]$	0.00459 $[0.0820]$	0.0671 $[0.165]$	0.0828 [0.117]	-0.00346 $[0.104]$	0.0633 $[0.117]$	0.160 $[0.126]$
Observations	11,736	17,338	17,338	3,343	17,338	17,338	14,483	17,338	17,338	14,041	17,183
Number of municipalities	978	1,099	1,099	213	1,099	1,099	915	1,099	1,099	1,099	1,099
Final year of sample period	2004	2008	2008	2008	2008	2008	2008	2008	2008	2008	2008
Conflict data source D/Come anome) Voor FF	NO No	CEDE Vac	CEDE	CEDE	CEDE	CEDE	CEDE	CEDE	CEDE	CEDE	CEDE No
(Distance to nearest border), x Year FE	No	No	Yes	No	No	No	No	No	No	No	No
Only municipalities near any border	No	No	$N_{O}$	Yes	No	No	$N_{O}$	$N_{O}$	$N_{O}$	No	No
$(FARC Events)_{i,1998} \times Year FE$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	Yes	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$
Border/One-from-border quadratic trend	$N_{O}$	$N_{O}$	$N_{O}$	No	$N_{O}$	Yes	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$
Estimation method	OLS	OLS	OLS	OLS	OLS	OLS	NB	OLS	OLS	OLS	OLS
$D(Border/One-from-border)_i \ge (Venezuela GDP)_t$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	$N_{O}$	Yes	$N_{O}$	$N_{O}$	$N_{O}$
$D(Border/One-from-border)_i \ge (Oil price)_t$	No	No	$N_{O}$	No	No	$N_{O}$	$N_{O}$	$N_{O}$	Yes	$N_{O}$	$N_{O}$
$(Department GDP)_{i,t}$	$N_{O}$	$N_{O}$	$N_{O}$	No	$N_{O}$	$N_{O}$	$N_{O}$	$N_{0}$	$N_{O}$	Yes	$N_{O}$
Local politics controls	$N_{O}$	$N_{O}$	$N_{O}$	No	$N_{O}$	$N_{O}$	$N_{O}$	$N_{0}$	$N_{O}$	$N_{O}$	$\mathbf{Yes}$
Notes: The dependent variable in column 1 is the $\overline{s}$ divided by 1993 nonulation. In all others (excent co	standardized	sum of uni the standa	lateral attac rdized sum	ks, massac of 19 indice	res and poli tors of FAF	tical kidnaj 3C activity	ppings by g divided bv	uerrilla grou 1993 nomils	tps, from Du	the and Var mn 7. "FAF	gas (2013), 3C Events"
is not divided by population nor standardized. All	regressions (	except colu	mn 7) includ	le municipa	ality fixed e	ffects, regio	n-year fixed	l effects and	a full set of	year intera	ctions with
fixed municipality characteristics. Column 7 includ-	les municipal	ity and yea	r fixed effect	ts. Column	s 2, 3 and 5	include as	additional	controls yea	arly interact	ions with a	dummy for
presence of coca crops in 2000, the minimum dista	ance to the 1	nearest bor	ler (km) an	d the value	of the del	bendent var	iable in 19	98, respectiv	/ely. The sa	mple in colu	umn 4 only
Column 6 includes a quadratic time trend specific	to border a	enezueia, u nd one-fron	таzи, геги, n-border mu	nicipalities	. In column	anuc anu 1 1 7, I use tl	he negative	binomial n	i numetuate iodel. Colun	ins 8 and 9	include as
additional controls the interactions of the border a	nd one-from-	border dum	mies with V	Venezuela's	GDP grow	th rate and	the price c	of oil, respec	tively. Colur	nn 10 inclue	les log real
GDP of the department where the municipality is k candidate in the last presidential election. Also inclu	ocated. Colu uded are seps	mn 11 inclu trate dumm	des as addit ies for Liber	ional contre al and Con	ols the vote servative m	share for th avors. Stanc	ie winning d lard errors	candidate aı clustered by	nd for the m	ost importaı v and depar	it left-wing tment-vear
(bootstrapped in column 7) in brackets. $*** p<0.0$ .	1, ** p<0.05	, * p<0.1.				0		6	4	4	0

 Table 4: Robustness checks

	(1)	(2)	(3)	(4)	(5)	(6)	(7)
	AUC	FARC	FARC	Armed Forces	FARC	FARC	FARC
VARIABLES	Events	Events	Events	Events	Events	Events	Events
$D(Chávez)_t \ge D(Border Venezuela)_i$	-0.0825	$0.328^{***}$	$0.304^{***}$	0.109	$0.267^{***}$	$0.303^{***}$	$0.207^{*}$
	[0.141]	[0.0983]	[0.0956]	[0.0879]	[0.0817]	[0.100]	[0.119]
$D(Chávez)_t \ge D(One-from-border Venezuela)_i$	0.150	0.105	0.106	-0.0616	0.156	0.192*	0.0261
	[0.126]	[0.132]	[0.133]	[0.0986]	[0.0952]	[0.104]	[0.107]
$(AUC Events)_{it}$		0.0944					
( )*)*		[0.0799]					
(Armed Forces Events) <sub>i</sub>		[]			0.520***		
					[0.0648]		
					[]		
Observations	17,319	17,319	17,338	17,328	17,328	17,338	17,338
Number of municipalities	1,099	1,099	1,099	1,099	1,099	1,099	1,099
$D(Any AUC Events)_i \ge Year FE$	No	No	Yes	No	No	No	No
$D(Border/One-from-border)_t \propto (FARC losses)_t$	No	No	No	No	No	Yes	No
$D(Border/One-from-border)_i \ge (US aid)_i$	No	No	No	No	No	No	Yes

# Table 5: The Chávez administration and counterinsurgent activity near the border with Venezuela

Standard errors clustered by municipality and department-year in brackets. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1. All regressions include municipality and region-year fixed effects. They also include a full set of year interactions with fixed municipality characteristics. The dependent variable in column 4 is the standardized sum of 9 indicators of activity for the Colombian Armed Forces divided by 1993 population. In the other columns, the dependent variable is the standardized sum of 19 indicators of activity for the corresponding group (FARC or AUC) divided by 1993 population. Column 4 includes as additional controls yearly interactions with a dummy for positive values of "AUC Events" at any point in the sample period. Columns 6 and 7 include interactions of the Border and One-from-border dummies with the aggregate yearly number of FARC losses (demobilizations + captures) and the amount of aid from the US government (in millions of 2011 dollars), respectively.